\documentclass[letterpaper,12pt]{article}
\usepackage[utf8x]{inputenc}
\usepackage{amsmath}
\usepackage[letterpaper,margin=1in,includehead=true]{geometry}
\usepackage{comment}
\usepackage{graphicx}
\usepackage{fancyhdr}
\usepackage{enumerate}
\pagestyle{fancy}
\usepackage{color}
\usepackage{wrapfig}
%\usepackage{enumitem}
\usepackage{tikz, tikz-3dplot, pgfplots}
\usetikzlibrary{calc}
\pgfplotsset{compat=newest}



\title{\vspace*{-1.5in}\textbf{Math 2300-007: Quiz 5a}}
\date{}
\author{}


\begin{document}
\maketitle
%\underline{Math 1300-400: Quiz 1}
\lhead{Math 2300-007: Calculus II}
\chead{Due: Feb., 16 2018}
\rhead{Quiz 5a}
\vspace{-.5in}
\noindent
{Name: \underline{\phantom{Aalsdkjflaskdjflaskjflksdjflkdsjf}}} \hfill{Score: \underline{\phantom{lksdjflkdsjf}}}
\vskip1cm
\noindent
This quiz is worth 5 points and is due at the beginning of class on \textbf{Friday, February 16, 2018}. Pick a couple of problems from Exam I for which you didn't recieve full credit. For each problem, write up a correct solution that includes a brief discussion of what went wrong, and how you could succeed on this kind of problem in the future. Making mistakes and learning from them is an integral component of getting better at math, so treat this as an opportunity for you to reflect on what you've learned and improve in the areas where you need some practice. Quality is more important than quantity here. Please staple this cover sheet to your write-up.\\


\noindent
Note: Quiz 5b will be an in-class survey on Friday, February 16 about how Math 2300 is going and what we can do during the rest of the semester to bolster our class community and maximize the learning therein. You will receive 5 points if you submit the survey.\\

\noindent
P.S. Use the white space below however you wish!
\end{document}
